Author's response to reviews

Title: Hispanic physicians' tobacco intervention practices: a cross-sectional survey study

Authors:

Francisco G Soto Mas (fsotomas@hsc.unt.edu)
Richard L Papenfuss (dpapenfuss@aol.com)
Holly E Jacobson (hollyej@flash.net)
Chiehwen Ed Hsu (edhsu@umd.edu)
Ximena Urrutia-Rojas (xurrutia@hsc.unt.edu)
William M Kane (kane@unm.edu)

Version: 6 Date: 8 November 2005

Author's response to reviews:

Formatting Changes

Thank you for your guidance. We have made the following changes:

Title & Abstract: These are identical in the manuscript and in the submission system

Author list: They are exactly as they are in the submission system

Figure and table placement text: We have deleted the placement text in the manuscript

Footnote: We incorporated the footnote into the text, pg. 9.

Author's contributions: The paragraph was reformatted as a single paragraph.

Figures: The format was changed to PowerPoint, and the white space minimize.